**Author's response to reviews**

**Title:** Antisickling and toxicological evaluation of the leaves of *Scoparia dulcis* Linn (Scrophulariaceae).

**Authors:**

Tavs A Abere (eseabere@yahoo.com)
Chiagozie J Okoye (chiagoziejoy@gmail.com)
Freddy O Agoreyo (agoreyofo@yahoo.com)
Gerald I Eze (justjerryelisa@yahoo.com)
Rose I Jesuorobo (rozy4god@gmail.com)
Clement O Egharevba (clement.egharevba@uniben.edu)
Pauline O Aimiator (aimiatorpauline@gmail.com)

**Version:** 5  
**Date:** 13 October 2015

**Author's response to reviews:** see over
Dear Editor,

We write to acknowledge the positive responses from the two Reviewers to the additions made to our manuscript.

In line with REVIEWER 1: Olubanke O Ogunlana’s recommendation, we have deleted “could be considered safe on acute basis”.

For REVIEWER 2: Jun-Won Yun, pharmacokinetics study was not factored into the original design of the work. We will recruit an expert in Pharmacokinetics into our Research team. The need to include other important tissues in histopathological examination is also noted.

Looking forward to hearing from you soon.

Thanks.

Dr. T. A. ABERE

Corresponding Author